Author's response to reviews

Title: Is reporting on interventions a weak link in understanding how and why they work: a preliminary exploration using community heart health exemplars

Authors:

Barbara L Riley (briley@healthy.uwaterloo.ca)
JoAnne MacDonald (jmacd069@uottawa.ca)
Omaima Mansi (omaima.mansi@mcgill.ca)
Anita Kothari (akothari@uwo.ca)
Donna Kurtz (donna.kurtz@ubc.ca)
Linda I VonTettenborn (vontettenbornl@douglas.bc.ca)
Nancy C Edwards (nedwards@uottawa.ca)

Version: 4 Date: 6 May 2008

Author's response to reviews:

As requested, the following format revisions were made:
- website was removed from the text and put in the reference list
- removed repeat headers from tables spanning 2 pages
- cropped the figure
- added a statement that all authors read and approved the final version

In addition to these requests, some very minor editorial changes were made, and affiliations of two authors were revised.

Thank you, once again, for your guidance and patience. Also, thank you for your leadership in building the field of implementation science.

Sincerely,

Barb Riley